Author's response to reviews

Title: Posttransplant Lymphoproliferative Disorder Involving the Ovary as an Initial Manifestation: a case report

Authors:

Takamitsu Inoue (takamitu@doc.med.akita-u.ac.jp)
Shigeru Satoh (shigerus@doc.med.akita-u.ac.jp)
Mitsuru Saito (mitsaito@med.akita-u.ac.jp)
Yohei Horikawa (horikawayo@gmail.com)
Norihiro Tsuchiya (tsuchiya@med.akita-u.ac.jp)
Tomonori Habuchi (thabuchi@doc.med.akita-u.ac.jp)

Version: 3 Date: 14 March 2009

Author's response to reviews: see over
March 13, 2009

Professor Michael Kidd, MD
Editor-in-Chief
JOURNAL OF MEDICAL CASE REPORTS

Dear Professor Kidd,

Thank you very much for accepting the manuscript entitled “Posttransplant Lymphoproliferative Disorder Involving the Ovary as an Initial Manifestation: a case report” (manuscript ID: 1783268781231399). I am herein sending the revised manuscript according to the formatting instructions.

Comment (1): Please include information on patient’s ethnic origin and country in the case presentation section.

   Re: We added “Japanese” and “whose ethnic origin was Asian” to the first sentence of the case presentation.

In addition, you send the image of only Figure 2b instead of whole Figure 2a-c although you did not mention the deletion of Figures. If you want to delete two of three CT images of Figure 2, please keep Figure 2a rather than 2b and delete the indications of Figure 2b and 2c in the manuscript (page 4, line 11 and page 5 line 1).

We would like to thank you for your helpful comments. We hope that the modified manuscript is now suitable for publication in JOURNAL OF MEDICAL CASE REPORTS.

Sincerely yours,

Takamitsu Inoue, M.D.
Department of Urology Akita University School of Medicine
1-1-1 Hondo, Akita 010-8543, Japan
Phone: 81-18-884-6156
Fax: 81-18-836-2619
e-mail: takamitu@doc.med.akita-u.ac.jp
takmitz@gmail.com